Author's response to reviews

Title: World Health Organization fracture risk assessment tool in the assessment of fractures after falls in hospital

Authors:

   Shin-ichi Toyabe (toyabe@med.niigata-u.ac.jp)

Version: 5 Date: 21 April 2010

Author's response to reviews: see over
The BMC Series Editorial Production Team,

April 21, 2010

Author list: Shin-ichi Toyabe

Title: World Health Organization fracture risk assessment tool in the assessment of fractures after falls in hospital

Journal: BMC Health Services Research

MS ID: 4102538282896851

Dear Editorial Production Team,

Thank you very much for informing me about problems of layout and format of my manuscript. I have corrected the problems that you have pointed out.

Sincerely yours,

Shin-ichi Toyabe, MD.

Crisis Management Office,
Niigata University Medical and Dental Hospital,
Asahimachi-Dori 1-754, Chuo-Ku, Niigata City 951-8520, Japan
TEL: +81-25-227-0375
FAX: +81-25-227-0375
E-mail: toyabe@med.niigata-u.ac.jp